



# DELIVERABLE

Project title:	LetItFlow: Active Distributed Workflow System for elderly
Project reference number:	AAL-2013-6-128

## D6.1 / D6.2 – Functional market analysis / Market Segmentation Report

**Revision: 0.1** 

Main Authors:

Pedro A. Ruiz (INTEGRASYS)

**Miguel Baizan (INTEGRASYS)** 

Ben Loke (Noldus)

Florina Sevescu (SIVECO)

Simona Bica (SIVECO)

Otilia Bularca (SIVECO)

Athena Mergeani (UHB)

Florina Antochi (UHB)

Date: [16/05/2016]
Dissemination Level: Public





# **REVISION HISTORY**

Revision	Author	Organisation	Description
0.1	Pedro A. Ruiz, Miguel Baizan	Integrasys	The first draft of the deliverable. Table of contents. Responsibilities assigned.
0.2	Pedro A. Ruiz, Miguel Baizan Otilia Bularca, Simona Bica	Integrasys SIVECO	First round of contributions from Integrasys. First round of contributions from SIVECO.
0.3	Pedro A. Ruiz, Miguel Baizan	Integrasys SIVECO	Update table of contents (merging deliverables D6.1/D6.2).
0.3 (1)	Florina Sevescu	SIVECO	Second round of contributions from SIVECO.
0.3 (2)	Ben Loke	Noldus	First round of contributions from Noldus.
0.3 (3)	Otilia Bularca	SIVECO	First review of the deliverable.
0.4	All	All	Content to the all sections.
0.5	Simona Bica, Athena Mergeani	SIVECO, UHB	Review and final contribution from SIVECO and UHB.





# **TABLE OF CONTENTS**

LIS	T OF	FIGU	JRES	5
LIS	T OF	TAB	LES	3
AB	BREV	<b>IATI</b>	ONS	3
1.	INT	ROD	UCTION	Э
1	.1	Sco	pe and objectives of the deliverable	Э
1	.2	Stru	cture of the deliverable	Э
1	.3	Con	sortium, project and product description10	)
1	.4	Met	hodology for market analysis1	1
1	.5	Mar	ket description12	2
	1.5.	1	Main market12	2
	1.5.	2	Alternative markets	7
2.	Exte	ernal	environmental analysis18	3
2	2.1	Mac	ro environment analysis18	3
	2.1.	1	Political environment18	3
	2.1.	2	Economic environment	1
	2.1.	3	Social environment	1
	2.1.	4	Technological environment	4
	2.1.	5	Ethical environment	5
2	2.2	Micr	ro environment analysis	3
	2.2.	1	Threat of new entrants	3
	2.2.	2	Bargaining power of suppliers	3
	2.2.	3	Competitive rivalry27	7
	2.2.	4	Bargaining power of buyers27	7
	2.2.	5	Threat of substitutive products	3
3.	Inte	rnal e	environment analysis28	3
3	3.1	Res	ources	3
	3.1.	1	Financial28	3
	3.1.	2	Physical assets	3





	3.′	3.1.3 Human Resources				
Э	8.2	Cap	abilities			
	3.2	3.2.1 Management of partnerships				
	3.2	2.2	Sales and Marketing			
4.	Si	milar p	projects and products in the market32			
5.	SV	VOT a	inalysis			
6.	Int	terview	vs with experts			
7.	Ma	arket s	egmentation			
7	<b>.</b> 1	Seg	mentation by Demography42			
7	.2	Geo	ographic segmentation44			
7	7.3 Behavioural segmentation					
7	7.4 Segmentation by Psychographic45					
8.	Se	electec	segmentation market46			
9.	Ini	tial Ma	arket strategy			
ç	).1	Inte	grasys48			
ç	9.2 SIVECO					
ç	9.3 Noldus					
ç	9.4 UHB53					
ç	9.5 HUVM					
10.		Concl	usions			
11.	BIBLIOGRAPHY					





# **LIST OF FIGURES**

Figure 1 Initial design of LetItFlow application on smartphone	11
Figure 2 Populations and employment developments in the European Union	13
Figure 3 ICT devices & services relevant for LetItFlow project	14
Figure 4 "Longevity economy" around the world	15
Figure 5 Rate of investments in 50+ market	15
Figure 6 Areas of Digital Health Innovation	16
Figure 7 Healthy life years for women	19
Figure 8 Healthy life years for men	20
Figure 9 Different statistics about active elderly people	21
Figure 10 - Median age of population, EU-28, 2001–14 period	22
Figure 11 - Population pyramid, EU-28, comparing 1994 and 2014	22
Figure 12 - Population pyramid, EU-28, forecast from 2014 to 2080	23
Figure 13 - Smartphone penetration per capita in Europe from 2011 to 2018	24
Figure 14 - Smartphone use in different groups of age	25
Figure 15 Respondents' domain of activity	39
Figure 16 Existence of workflow management component in the IT system of the questioned organization	40
Figure 17 Staff monitoring tool existence	41
Figure 18 Usefulness of IT solutions for elders	42
Figure 19 Segmentation by Demography	43
Figure 20 Segmentation by geography	44
Figure 21 Segmentation by behavior	45
Figure 22 Segmentation by psychographics	46





# **LIST OF TABLES**

Table 1 Scopes and purpose of deliverable	9
Table 2 Conceptual approach of AAL in LetItFlow project	13
Table 3 Retirement ages in Europe	23
Table 4 Threat of new entrants	26
Table 5 Bargaining power of suppliers	27
Table 6 Competitive rivalry	27
Table 7 Bargaining power of buyers	27
Table 8 Threat of substitutive products	28
Table 9 - Physical assets of the LetItFlow partners	29
Table 10 – Technology and management expertise	30
Table 11 – Management of Partnerships	31
Table 12 – Sales and Marketing	32
Table 13 – Similar projects to LetItFlow	33
Table 14 – Similar products to LetItFlow	34
Table 15 SWOT, strengths	34
Table 16 SWOT, weaknesses	35
Table 17 SWOT, opportunities	35
Table 18 SWOT, threats	35
Table 19 Questionnaire	38
Table 20 Assisted group	47
Table 21 Productivity increase group	47
Table 22 Price sensitive group	48
Table 23 Initial Business Plan - Integrasys	50
Table 24 Initial Business Plan – Siveco	51
Table 25 Initial Business Plan – Noldus	53





Table 26 Initial Business Plan – UHB	54
Table 27 Initial Business Plan - HUVM	55





## **ABBREVIATIONS**

- AAL Ambient Assisted Living
- SWOT Strengths, Weaknesses, Opportunities and Threats
- PEST Political, Economic, Social and Technological
- AARP American Association of Retired Persons
- USA United States of America
- EU European Union
- GDB Gross Domestic Product
- IT Information and Technology
- ICT Information and Communication Technology
- US United States
- R&D Research and Development
- AAL JP Ambient Assisted Living Joint Program
- DG ECFIN Directorate General for Economic and Financial Affairs
- UHB Emergency University Hospital of Bucharest, Romania
- HUVM Virgen Macarena University Hospital Seville, Spain





# **1. INTRODUCTION**

#### **1.1** Scope and objectives of the deliverable

The scope of this deliverable is to do market research for the LetItFlow project, which mainly contains task 6.1 (Functional market analysis) and 6.2 (Market segmentation report). This deliverable is aimed at increasing market knowledge, especially the market about assisted working operations for elderly people. The data, information and literature used to complete the deliverable come from previous deliverables D2.1, D.2.2 and D2.3, desk research, interviews, internet and related books as well as literature.

This deliverable can be divided basically into two parts. The first part aims to get insight into the external and internal environment of the market for LetltFlow project; the second part focuses on market segmentation analysis according to end-users' requirements, as well as an initial strategy. The segmentation is based on the outcome of the user requirements report (deliverable 2.1) and the external and internal analysis. This second part was going to be covered separately in deliverable D6.2 (market segmentation report) although finally will be integrated all information together in the deliverable D6.1. This deliverable will be input for the business model (deliverable 6.3) and exploitation plan (deliverable 6.4).

Purpose	1.Increase market knowledge	2.Conduct external and internal analysis	3. Provide initial Strategy
Detailed Market Plan	a. Desk research b. Interview c. Survey	<ul> <li>a. Macro Environment Analysis (PEST)</li> <li>b. Industrial environment analysis.</li> <li>c. Competitive environment analysis.</li> <li>d. Stakeholder analysis.</li> <li>e. Internal strengths and weaknesses.</li> <li>f. SWOT Analysis.</li> </ul>	<ul> <li>a. Segment Market Analysis.</li> <li>b. Selected Segments.</li> <li>c. Input for Business Plan.</li> </ul>

The exact scopes and purpose of this deliverable are showed below:

 Table 1 Scopes and purpose of deliverable

External and internal analysis will contribute to our promotion strategies, distribution strategies and market segmentation. Based on this, possible segmentation and strategies are developed, which will help to access and develop the market for the outcome of this project.

#### **1.2 Structure of the deliverable**

This report is structured in a number of chapters that we will briefly comment below:

- Chapter 1 explains the plan, scope, objective and structure of the deliverable. Moreover, it introduces this project, product, main market and research questions.
- In chapter 2 the external environment is described based on PEST analysis.
- Chapter 3 presents the internal environment analysis.
- Chapter 4 provides similar projects and products.





- Chapter 5 formulates the SWOT analysis.
- Chapter 6 provides the results and conclusions of interviews.
- In chapter 7, the LetItFlow market is segmented and selected segments are described more elaborately.
- Chapter 8 provides several possible market strategies on the basis of segmented market.
- Chapter 9 gives an overview of the market strategy and positioning, also with regards to competitors.
- Chapter 10 gives the final conclusion.

#### **1.3** Consortium, project and product description

With the increase of aging in our society, the need for technological devices by elderly people is growing. At the same time, the progressive deterioration of physical and psychological abilities diminishes the capacity to perform their daily life activities. So elderly people would like to feel safer by using technological devices in their environment, for example smartphones, because such devices could help them to keep in touch with their children or ease the daily operations in the work.

In order to help solving this problem, the project LetItFlow is conducted by several organizations and it is made up of six technology and final users partners spread over the European continent: Spain: Integrasys (Spain – Leader), HUVM (Spain), Noldus (Netherlands), AIT (Austria), SIVECO (Romania), UHB (Romania).

The LetItFlow project aims to design and implement a workflow support system to help elderly hospital workers (mainly nurses) to accomplish their daily tasks and to the adaptation to new procedures and methods. The system should enable assistance by means of workers activity monitoring, portable communication tools, alerting services, adapted interaction interfaces or workflow management. Although we will particularize the solution to hospitals, the system could be applied to different work domains.

LetItFlow combines two challenges: change management in companies and workflow technologies for elderly workers, being the general objective to facilitate the adaptation of older workers to changeable work environment and to perform their tasks by supporting them with real-time context aware tool for guiding them in daily tasks.

This tool will have two types of users:

- The instructors: the users that create the instructions (or tasks) and monitor the employee to guide the worker in their workday.
- The employees or the instruction receivers: the employees themselves who benefit from the tools. As we comment previously, our case the employees are mainly nurses.

LetItFlow is based on fixed and mobile platforms that interact with the nurses to guide them in their work activities. Different Human-Computer interfaces are considered in order to adapt to the user activities and user profile (tactile, voice, text, vibration and multimedia). After an extensive search no similar solution in the market have been found, although we will build it from existing technologies (Hospital Information System and Workflows implementation solution from SIVECO, user behaviour and tracking from NOLDUS, Context monitoring and communication framework from Integrasys and user-friendly and adapted interfaces from AIT), products and standards that are helping to the adoption of the platform by the market. The lack of this type of framework and tools, together with the expertise of the consortium and potential market impact, will suppose incremental innovations as the approach is trying to use existing solutions as much as possible.





After analyzing the end-user requirements of the LetItFlow system through online surveys, focus groups, literature and project research and experiences from previous projects, the specific functional and non-functional requirements were developed from the data collected at two distinguished hospitals from Romania and Spain that are partners in the consortium. Finally, one proposed solution is formulated in base to these requirements where we have basically a mobile application on the basis of a smartphone or tablet and its final benefit is supporting elderly people to perform their works independently and help them to feel confident and safe while increasing the efficiency of their work.



Figure 1 Initial design of LetItFlow application on smartphone

#### 1.4 Methodology for market analysis

A market analysis for a product or business is one of the most basic and useful tool for strategic business planning, basically it is an analysis that provides a straightforward model with strengths, weaknesses, opportunities and threats to create the foundation of a marketing strategy. Definitions of strategic analysis often differ, but the following attributes are commonly associated with it:

- 1. Identification and evaluation of data relevant to strategy formulation.
- 2. Definition of the external and internal environment to be analysed.
- 3. A range of analytical methods that can be employed in the analysis.

The analytical methods used in this market analysis include:

- SWOT for Internal Analysis. SWOT analysis is an acronym for strengths, weaknesses, opportunities, and threats and is a structured planning method that evaluates those four elements of a project or business venture. It involves specifying the objective of the business venture or project and identifying the internal and external factors that are favorable and unfavorable to achieve that objective.
- PEST for External Analysis. PEST analysis is a scan of the external macro-environment in which an organisation exists. It is a part of the external analysis when conducting a strategic analysis or doing market research, and gives an overview of the different macroenvironmental factors that the company has to take into consideration. It is a useful strategic tool for understanding market growth or decline, business position, potential and direction for operations.
- Porter for Competitiveness Analysis. Porter's five forces analysis is a framework that attempts to analyse the level of competition within an industry and business strategy development. It draws upon industrial organization economics to derive five forces that determine the competitive intensity and therefore attractiveness of an Industry.





#### **1.5 Market description**

#### 1.5.1 Main market

In this chapter we're going to analyse the target group of LetItFlow, taking into consideration the impact of ageing population over the long-run economic and budgetary projections.

LetItFlow solution is shaped around the specific workflows and assists elderly to perform their daily work tasks using various ICT communication tools. This solution is destined to be used by employees over 50 years old and is not limited by activity domain, country or region. LetItFlow is tailored to the needs, expectations, lifestyle, preferences and routines of the individual but at the same time, is focused on social, political and legal systems and technological environment.

Solutions like LetItFlow are relevant nowadays because worldwide population is growing old and average life and the expectancy has increased from 55 years in 1920 to over 80 years today. The Ageing Report<sup>1</sup> issued by The European Commission stated that "one in three Europeans will be over 65 by 2060. This implies that the EU would move from having four working-age people for every person aged over 65 years to about two working-age persons. So, aiming to keep elderly working and shaping the working environment so they can be more efficient and satisfied, becomes a desideratum. Even if future generations will be more technology-literate, there will always be people ready to try anything new and people who have difficulties or do not want to adapt to new media and devices.

For the EU market, total government spending on pensions, healthcare, long-term care, unemployment benefits and education will increase by almost 20% (or 4.1 percentage points of GDP), while expenditures for long-term care will double, between 2010-2060. The long-term budgetary projections show that population ageing poses a challenge for the public finances in the EU. The fiscal impact of ageing is projected to be high in most Member States, with effects becoming apparent already during the next decade.

Based on the same report, labour force projections show a rise in overall participation rates, particularly visible for ages 50+, reflecting the combined effect of the rising attachment of younger generations of women to the labour market, together with the expected impact of pension reforms. By large, the biggest increases in participation rates are projected for older workers (around 21 pp. for women and 10 pp. for men) in the EU for the age group 55-64, influenced by enacted pension reforms. Consequently, the gender gap is projected to narrow substantially in the period up to 2060. The total participation rate (for the age group 20-64) in the EU is projected to increase from 76.5% in 2013 to 80.1% in 2060.

On the same note, demographic developments have a major impact on labour market developments. Three distinct periods can be observed for the EU as a whole:

- 2007-2011 demographic developments still supportive of growth: the working-age population is growing, but employment is sluggish as the financial and economic crisis weighs on labour prospects during this period.
- 2012-2022 rising employment rates offset the decline in the working-age population: the working age population starts to decline as the baby-boom generation enters retirement. However, the assumed reduction in unemployment rates, the projected increase in the

<sup>&</sup>lt;sup>1</sup> http://ec.europa.eu/economy\_finance/publications/european\_economy/2015/pdf/ee3\_en.pdf





employment rates of women and older workers cushion the impact of demographic change, and the overall number of persons employed would start to increase during this period.

• From 2023 – the population ageing effect dominates: the projected increase in employment rates is slower, as trend increases in female employment and the impact of pension reforms will be less pronounced. Hence, both the working-age population and the number of persons employed start falling over the remainder of the period.



#### Figure 2 Populations and employment developments in the European Union

Balancing the health care needs of the European population with spending resources, as well as continuous efforts to increase the efficiency and quality of health service delivery, will continue to be high on the political and economic reform agenda of Member States.

In one word "ageing well" becomes a desideratum of this era and IT solutions gain more trust when it comes about assisting older adults in their daily approaches. In this context a series of sectors have relevance. Ambient Assisted Living Program targets older adults (over 55 year old, pension-ready and freshly retired persons). Those persons are to be found more and more in large activity areas like enterprise, business, industries or public institutions. All those domains meet and tend to continue their "life" under this program that becomes a market of markets.

Efficiency and well-being (personal comfort)	Health care and monitoring
Remain engaged at their workplace as much as possible	Improve management of chronic conditions
Security and safety	Personal empowerment and quality of life

#### Table 2 Conceptual approach of AAL in LetItFlow project





Together these sectors comprise a rather fragmented socio-technical environment for the deployment of a potentially very wide range of ICT-based solutions. Within LetItFlow project the ICT "offer" comprises of: Services, Devices and IT tools.



Figure 3 ICT devices & services relevant for LetItFlow project

User-friendliness and accessibility is the heart of LetItFlow. When developing LetItFlow are taken into consideration: personal constraints related to ageing, such as gradual loss of sensory abilities (hearing, eyesight etc.), physical impairments (speech, dexterity, mobility, strength, endurance, etc.) and cognitive and intellectual problems (memory loss, information processing, etc.) impact on the interest and ability to engage with new technologies and the confidence in using them.

Over the ocean, in The United States of America the impact of ageing population is similar to Europe's situation.

The HEALTH INNOVATION FRONTIERS report<sup>2</sup> published by AARP in 2014 emphasises the importance of 50+ "longevity economy" as the third largest in the world. In fact, only the United States and China are larger. The medical system's efforts traditionally are oriented towards sickness rather than health, and treatment rather than prevention - thus leaving a 'healthy living' gap. Large opportunities exist for new and meaningful investment aimed at meeting healthy living needs of all ages.

<sup>&</sup>lt;sup>2</sup> https://ec.europa.eu/research/innovation-union/pdf/active-healthy-ageing/aarp.pdf







#### Figure 4 "Longevity economy" around the world

In United States of America, productive older workers, second-careerists and retirees will within 17 years account for more than half of the its gross domestic product. Older Americans will soon outnumber children, not only because of population growth but because they are living longer.

Investments related to the 50+ market continue to grow at the pace of investments in the larger digital health and wellness market.



#### Figure 5 Rate of investments in 50+ market

This huge market is in many ways untapped, and the priorities of this massive population are often unaddressed by market players. Opportunity estimates to serve the 50+ with Health and Wellness products/services to generate \$30 billion in cumulative revenue over the next five years (2014-19), across nine areas of Digital Health Innovation.



Figure 6 Areas of Digital Health Innovation

In view of solution architecture, Gartner analysts predict that in 2016 "Algorithms Shift the Focus to Business-and Human-Centric Emerging Technologies"<sup>3</sup> in the report published on 3<sup>rh</sup> December 2015. Algorithms are becoming even more relevant today, because of the pervasiveness of highly connected devices and things, and the ripple effect this has on how organizations can use, scale and extend algorithms to increase their revenue.

Emerging technologies become more human-centric and reflect a significant shift in how humans and technology interact; emerging technologies reflect human behaviours, rather than humans adapting to technology. This results in humans creating unique and new relationships with their technology.

The focus shifts from technology to the information. Traditionally, technology has been a primary focal point, but not anymore. Since the majority of the technologies used to enable ambient digital experiences have existed and have been maturing for over a decade, the focus shifts to how edge technologies will generate new and meaningful insights with the enormous amount of information being generated.

We can conclude that there is a real market for LetItFlow solution in Europe and also in America and it is growing continuously. Moreover, adapting our solution to user needs, we can state that it a human-centric solution and is in line with IT industry trends.

#### Actors (key players)

Organisations and institutions that:

<sup>&</sup>lt;sup>3</sup> https://www.gartner.com/doc/3174921/predicts--algorithms-shift-focus





- Have a high rate of employees over 55 years.
- Belong to activity domains with lack of qualified personnel.
- Belong to activity domains with stress and workload (work in critical conditions).

#### 1.5.2 Alternative markets

LetItFlow is a robust, attractive and customizable solution for various domains of activity. In this chapter, we're going to describe how the modules developed in LetItFlow can be used in other markets.

LetItFlow is targeted to **elderly workers from hospital and laboratory (healthcare domain)**. However, as a side effect of the project requirements, other type of users might benefit from the LetItFlow solution. For instance the LetTrain application may serve to train **newcomer workers** and accelerate the learning process by offering an interactive guide about the different processes involved in the hospital and laboratory. While in this case the main market is still the same, by increasing the number of users' profile, new market opportunities are opened.

**Guard and protection** organizations need to have very fast solutions to transmit alarms/alerts in case of life threatening situations. Examples could be various but for each one, knowing the sender's location is essential. A guard should have a smart watch or smart phone to send the alarm using LetAlarm module. This module can be used in a similar way by **geophysicists** who deal with seismology and meteorological hazards and need to send alarm messages right away.

**Water purification stations** have strictly workflow processes as disinfection, fluoridation or filtration. Employees who work in this environment have a great deal of responsibility and need to have a lot of experience even though many processes are semi-automated. LetApp would guide them during their work day and they can check what was done and share this information with team members. LetAlarm could also be useful in case dangerous situations arise. New workers can be guided at first using LetTrain.

Other domains with strict activity processes as **chemistry/biochemistry**, **nuclear physics/isotopes**, and **waste management** can use these modules in a similar way. In all cases human error is possible and sometimes can have devastating results if for example one skips or switch steps during a process. Users could access LetTrain training content whenever necessary. Because these domains have high risks, is important to record vital signs of key users with health problems using LetCritical and when vital signs are hitting critical points it will send an alert/ notification to LetAlarm.

**IT companies** (IT / applied IT) can benefit of LetItFlow modules by assigning tasks with LetApp or by providing access to training content with LetTrain.

Identification of user location in the building/area can be of interest and it's done with LetAlarm.

For Noldus, the LetCritical application that is developed for monitoring the health condition of nurses, which in combination with localization services, could also be applicable in other market segments. At this stage we did not yet explore the opportunities in great detail, but there are more human behaviour research applications in which the behaviour of persons in combination with their location is relevant. Examples can be found in the area of consumer behaviour research, human factors and psychology.

Nowadays, business process framework and workflow engines are tools widely used in diverse industrial environments and with different scopes (i.e. not only hospitals, but any markets which needs to model business processes in the general sense). Even though the processes modelled in LetItFlow will be surely not adequate for these other markets, the technological infrastructure





created together with the knowledge acquired in the project open a wide range of possibilities to develop new processes and facilitate the entrance to other markets. Some pieces of the LetItFlow solution might be reused in those cases (e.g. users' management, task assignation procedures, workflows visualization, etc.).

## 2. EXTERNAL ENVIRONMENTAL ANALYSIS

#### 2.1 Macro environment analysis

Normally, macro environment is analyzed using the PEST method, which stands for the political, economic, social (including legal and cultural) and technological environment. Additionally the ethical factors are included.

#### 2.1.1 Political environment

Political factors about how the government intervenes in the economy and the market relative to our product.

#### Local governments encourages R&D on health sector

The R&D investments in health generate each year tangible and positive outcomes for us today. The most self-evident direct benefit of investing in health R&D is the subsequent improvement in health outcomes and longevity of people what provokes a good opportunity of market for the next years. On the other hand, increased investment in R&D has also a fundamental role to play in economic growth in Europe as there are direct and indirect links between increasing R&D spend on healthcare, improved healthcare services, and the consequent wider benefits to the overall economy. An example of these investments by Local European governments is the AAL programme where this project is located.

#### *Europe encourages to develop ICT for supporting the elderly*

Information and Communication Technologies applied to health and healthcare systems can increase their efficiency, improve quality of life and unlock innovation in health markets. During the period 2014-2020, research and innovation will be supported under "Health, demographic change and wellbeing" of Horizon 2020, in the areas of:

- An ICT and computational science and engineering framework for digital, personalised, and predictive medicine, including advanced modelling and simulation.
- Innovative instruments, tools and methods for unlocking the value of data and for advanced analytics, diagnostics and decision making.
- New digital media, web and mobile technologies and applications, as well as digital instruments that integrate healthcare and social care systems and support health promotion and prevention.
- eHealth systems and services with strong user involvement, focusing on interoperability and the integration of emerging patient-centric technologies for cost-effective healthcare.

The AAL programme is the one of best examples of the interests in Europe to support elderly through ICT. Specifically, the programme is the funding activity that aims to create better conditions of life for the older adults and to strengthen the industrial opportunities in Europe through the use of information and communication technology. The AAL JP programme has invested significantly in





almost 130 projects since 2008 and several of them are now beginning to show results and demonstrating real market potential.

Dealing with demographic changes is also one of the focus points in the Horizon2020 call, which is another example showing that European governments do see the urgency, and is willing to invest in related developments.

#### *Healthy life years*

Population ageing is one of the greatest social and economic challenges facing the EU. Projections foresee a growing number and share of elderly persons (aged 65 and over), with a particularly rapid increase in the number of very old persons (aged 85 and over). These demographic developments are likely to have a considerable impact on a wide range of policy areas: most directly with respect to the different health and care requirements of the elderly, but also with respect to labor markets, social security and pension systems, economic fortunes, as well as government finances.

In 2013, the number of healthy life years at birth was estimated at 61.4 years for men and 61.5 years for women in the EU-28; this represented approximately 79 % and 74 % of total life expectancy for men and women.

The next charts from Eurostat show the healthy life years for women and men comparing the years 2010 and 2013, and where we can see how healthy living is increasing.



(\*) Estimates. (\*) 2010: not comparable due to a break in series. (\*) Data for 2011 instead of 2010. (\*) Data for 2012 instead of 2013. Source: Eurostat (online data code: htth\_hlye)

Figure 7 Healthy life years for women







(\*) Estimates. (\*) 2010: not comparable due to a break in series. (\*) Data for 2011 instead of 2010. (\*) Data for 2012 instead of 2013. Source: Eurostat (online data code: htth\_hlye)

#### Figure 8 Healthy life years for men

#### Innovation for Active and Healthy Ageing, from concept to reality, from local to global

Demographic change opens a wide variety of new opportunities for European innovators and industrial players to develop new products and services, tailored to the needs of a growing number of ageing consumers. For the above, the governments are increasing the funding and the investing with different programs to promote to the companies the investigation in this group of population. Technological and systemic innovation together can increase the efficiency of our health and care systems, improve the quality of life of the older population and, at the same time, create new opportunities for innovators.

#### Ageing population political priority

Vastly improved life expectancy, one of the great triumphs of the last century, looks set to be one of great challenges of this one. Between 2015 and 2020, over a period when the general population is expected to rise 3%, the numbers aged over 65 are expected to increase by 12% (1.1 million); the numbers aged over 85 by 18% (300,000); and the number of centenarians by 40% (7,000).

Much of the costs of old age have arisen because growth in total life expectancy has outpaced growth in healthy life expectancy (i.e. the number of years we can expect to live in good health). Policies that improve preventative healthcare, and help people to remain active and healthy in later life, could help increase the proportion of life spent in good health and reduce costs.

A healthier old-age population would also allow greater numbers to remain in the labour market for longer, thereby mitigating the impact of an ageing population on the dependency ratio.







Figure 9 Different statistics about active elderly people

#### 2.1.2 Economic environment

Economic factors include economic growth, interest rates, exchange rates and the inflation rate. These factors greatly affect how businesses operate and make decisions.

#### DG ECFIN reports

DG ECFIN produces economic forecasts on behalf of the European Commission. DG ECFIN's forecasts are published three times a year in sync with the EU's annual cycle of economic surveillance procedures, known as the European Semester. At the time of writing the current document, the autumn 2015 European Economic Forecast, published on 5 November 2015, is the last forecast released. According to this report, the economic recovery in the euro area and the European Union as a whole is now in its third year. It should continue at a modest pace next year despite more challenging conditions in the global economy. Within the EU, the expected growth of GDP in 2016 is approximately of 2%.

#### 2.1.3 Social environment

Social factors include the cultural aspects and health consciousness, population growth rate, age distribution, career attitudes and emphasis on safety.

#### The number of elderly people is increasing

The number of people aged 65 and above is increasing in many countries. Consistently low birth rates and higher life expectancy will transform the shape of the EU age pyramid. The ageing society will result in an increasing demand for health care services, while at the same time the number of health care workers will decrease.

The following figures, published by Eurostat, represent this tendency:







(\*) 2010, 2011, 2012 and 2014: break in series. 2013: provisional. 2014: provisional; estimate. Source: Eurostat (online data code: demo\_pjanind)



Figure 10 - Median age of population, EU-28, 2001–14 period

(1) 1994: EU-27. 2014: provisional; estimate. Source: Eurostat (online data code: demo\_pjangroup)

Figure 11 - Population pyramid, EU-28, comparing 1994 and 2014







(\*) 2080: projections (EUROPOP2013). Source: Eurostat (online data codes: demo pjangroup and proj 13npms)

#### Figure 12 - Population pyramid, EU-28, forecast from 2014 to 2080

#### *The retirement age is increasing in some European countries*

As a consequence of what has been exposed in the previous section, some local governments are increasing the retirement age, as they see that elderly people is now in better health conditions to work as those of the same age a decade or two ago. An example of this fact is the Spanish government, who has recently increased this margin from 65 to 67 years.

Next table shows the current retirement age for the countries involved in the project and the projection to the next years.

EU	Current general retirement age Men/ Women	Future retirement age
Austria (AT)	65 / 60	65 (2033)
Romania (RO)	65 / 60	65 (2030)
Spain (ES)	67month	67 (2027)
Sweden (SE)	61-67; 65	_
Netherlands (NL)	65 and 3month	67+ (2024)

#### **Table 3 Retirement ages in Europe**

In the EU15 states, the general retirement age is 65 years. In most new Member States, the retirement age will be raised to the same level over the next decade. Denmark, France, Germany





and Spain have decided to increase the retirement age from 65 to 67 years. For the main part, the changes in retirement ages will occur in the 2020s.

#### 2.1.4 Technological environment

Technological factors include technological aspects like R&D activity, automation, technology incentives and the rate of technological change. These can determine barriers to entry, minimum efficient production level and influence the outsourcing decisions.

*Increased use of smartphone and tablet among elderly people* 

Despite new technologies have traditionally been a barrier for the elderly, the smart devices have grown increasingly common in recent years and are penetrating more and more into this population sector. Almost all statistics agree indicating that the penetration of the smartphones in people above 55 years is close to 60% and it is rapidly increasing. The increased penetration in society of inexpensive portable devices, sensors and cameras, enables the development of assistive tools at more affordable costs.

The statistic depicts the smartphone penetration per capita in Europe from 2011 to 2018:



Figure 13 - Smartphone penetration per capita in Europe from 2011 to 2018

#### Elderly workers more reluctant to adopt new technologies in the working environment

A commonly held belief is that older people are resistant to change and have negative attitudes toward the use of technology. However, the available data dispute this stereotype and indicate that, in general, older people are receptive to using technology if they perceive the technology as useful, if the technology is easy to use, and if they are provided with adequate training and support.





Apart from this, the technological advances are usually perceived as a threat for the job position, as in many cases it has meant a reduction on the work force and has derived in dismissals.

The following chart shows how much cell phone use differs between different age groups.



Figure 14 - Smartphone use in different groups of age

#### 2.1.5 Ethical environment

The ethical environment analysis provides a basis for what is right and what is not based on elderly people.

#### *Recording a location of people*

Recording and tracking the location of persons at the workplace is a sensible subject. Data protection law doesn't prevent monitoring in the workplace. However, it does set down rules about the circumstances and the way in which monitoring should be carried out. Before deciding whether to introduce monitoring, it is necessary to sufficiently inform the workers (reasons for monitoring, benefits that this will bring, identify negative effects, etc.). In general, apart from legislation, people are more aware of privacy and confidentiality of data.





#### Confidentiality of information

Compared to younger people, the information of elderly people is more easily to be leaked when they use ICT, because they have less experience and knowledge. Some older users may not choose the right way to use the service or may not be conscious about the openness of their information. In addition, some devices need personal information, like health status, to service the elderly better. Many patients want to employ digital and mobile technologies in their medical care. However, many of them are worried about the privacy and security of mobile devices and wonder whether increased reliance on handheld devices will compromise the confidentiality of their medical information. They fear loss of a smartphone or tablet that contains personal material and are concerned that their medical information might end up in the hands of employers or private individuals. Moreover, in many countries, neither public nor private insurance plans cover such applications. So security and privacy technology are very important for developing mobile devices for elderly (West, 2012)<sup>4</sup>.

#### Loss of personal autonomy

The inclusion of a workflow engine together with electronic devices in the workday of elderly workers can provoke a feeling of loss of personal autonomy and a feeling of rejection when using it due to lack of use in their daily lives.

#### 2.2 Micro environment analysis

Porter's Five Forces model identifies and analyses five competitive forces that shape every industry, and helps determine an industry's weaknesses and strengths. These forces are *Threat of new entrants, Bargaining power of suppliers, Competitive rivalry, Bargaining power of buyers, Threat of substitutive products.* 

Porter identified these five undeniable forces that play a part in shaping every market and industry in the world. The forces are frequently used to measure competition intensity, attractiveness and profitability of an industry or market.

#### 2.2.1 Threat of new entrants

A company's power is affected by the force of new entrants into its market. The less money and time it costs for a competitor to enter a company's market and be an effective competitor, the more a company's position may be significantly weakened.

Threat of new entrants		
Strength: Medium	<ul> <li>Entry barriers are relatively high due to high capital requirements/ Long payback period</li> <li>High scope on product Differentiation</li> <li>High government incentives</li> </ul>	

#### Table 4 Threat of new entrants

#### 2.2.2 Bargaining power of suppliers

<sup>&</sup>lt;sup>4</sup> http://www.insidepolitics.org/brookingsreports/mobile\_health\_52212.pdf





This force addresses how easily suppliers can drive up the price of goods and services. It is affected by the number of suppliers of key aspects of a good or service, how unique these aspects are and how much it would cost a company to switch from one supplier to another. The fewer number of suppliers, and the more a company depends upon a supplier, the more power a supplier holds.

Bargaining power of suppliers	
Strength: Low	<ul> <li>Suppliers: BPM solution providers, Mobile Devices suppliers, Server suppliers.</li> <li>We are committed to grow strong partnerships with suppliers to maintain a low bargain power.</li> </ul>

 Table 5 Bargaining power of suppliers

#### 2.2.3 Competitive rivalry

The importance of this force is the number of competitors and their ability to threaten a company. The larger the number of competitors, along with the number of equivalent products and services they offer, dictates the power of a company. Suppliers and buyers seek out a company's competition if they are unable to receive a suitable deal.

Competitive rivalry	
Strength: Low	<ul> <li>Nowadays are less differentiated products dedicated to older employees.</li> <li>In the next few years, growth opportunities arise worldwide due to governmental incentives and a genuine interest to help elders integrate better and be more productive at work.</li> </ul>

#### Table 6 Competitive rivalry

#### 2.2.4 Bargaining power of buyers

This specifically deals with the ability customers have to drive prices down. It is affected by how many buyers, or customers, a company has, how significant each customer is and how much it would cost a customer to switch from one company to another. The smaller and more powerful a client base, the more power it holds.

Bargaining power of buyers	
Strength: Medium	<ul> <li>Buyers   organizations that activate in the following domains: chemistry / biochemistry, nuclear physics, waste management, purification stations, geophysics, medical, guard and protection, IT / applied IT.</li> <li>Bargaining power of buyers is relative. This could be related to their lack of interest to adapt new solutions.</li> <li>Buyers are very price sensitive.</li> <li>High switching costs.</li> </ul>

 Table 7 Bargaining power of buyers





#### 2.2.5 Threat of substitutive products

Competitor substitutions that can be used in place of a company's products or services pose a threat. For example, if customers rely on a company to provide a tool or service that can be substituted with another tool or service or by performing the task manually and this substitution is fairly easy and of low cost, a company's power can be weakened.

Threat of substitutive products		
Strength: Low	<ul> <li>Preference for less customized solutions for elders</li> <li>Patent protection.</li> <li>Threat of other innovations.</li> </ul>	

 Table 8 Threat of substitutive products

## 3. INTERNAL ENVIRONMENT ANALYSIS

The resources and capabilities of project partners have decisive influence on market strategies. Resources will be analysed from financial, physical, human and technological perspectives while capabilities will focus on the management and marketing. We will conclude the section by listing the strengths and weaknesses of both, resources and capabilities.

#### 3.1 Resources

#### 3.1.1 Financial

Six project partners are involved in LetItFlow project. The exact financial resources of these organizations will, where relevant, be addressed in future deliverable 6.3 (Business models and exploitation).

#### 3.1.2 Physical assets

Table 9 enumerates the different physical resources of the LetItFlow partners related to plan, equipment, assets, technology and raw materials.

Partner	Physical Resources
Integrasys	Offices at two physical locations (Madrid and Seville). Hardware, computers, smart devices, software applications, manufacturing systems and test equipment for demonstration purposes.
Noldus	Headquarters in Wageningen (NL), where most of R&D development takes place. Additionally offices in China, US, Germany, France, Italy, Spain and Hungary, and a worldwide network of distributors. In addition observation labs, hardware, computers, smart devices, software applications, manufacturing systems and test equipment for demonstration purposes.



Τ

Γ



AIT	Headquartered in Vienna with several other locations in Austria. As Austria's biggest non-university research institution (more than 1200 employees), AIT can rely on substantial physical resources related to key infrastructure topics such as energy, mobility, safety & security, health & environment, and innovation systems. Most relevant for this project is the UX Laboratory based at the Technology Experience Business Unit, which is equipped with state-of-the-art technology for user research and demonstration activities.
SIVECO	SIVECO has four international offices (Brussels, Ankara, Astana, and Dubai). Most of its IT equipments are in the headquarter of Bucharest and include: computers, servers and smart devices.
UHB	University and Emergency Hospital of Bucharest (UHB) is a public health unit having a role in assuring medical services (emergency, prevention, recovery, and palliative care). UHB is composed of academic clinical and non-clinical units, which provide medical assistance, university activities, clinical research, and post-academic education having contractual relationships with approved medical faculties.
	UHB is since 2011 a Category 1 regional emergency hospital, with a high level of competency.
	The Research and Development Unit in the University and Emergency Hospital of Bucharest collaborates with national and international partners in order to perform clinical studies and clinical research. The research activity strengthens the quality of medical act, while guaranteeing patients' rights, morals and ethics. The Research Unit is currently involved in two international financed grants, 3 grants with internal financing and 183 multicentric international studies.
	The Neurology Research Unit has participated in the last 4 years in 7 grants with internal financing, more than 20 international multicentric clinical trials, one grant with external financing regarding genetic assessment of risk in stroke subtypes.
HUVM	It is placed in Seville and has assigned a population of 551856 inhabitants. Yearly they handle more than: 36000 admissions, 40000 operations, 800000 specialized consults, 64000 patients in day hospital, 11000 home assistance and 200000 attended urgencies.
	It consists of two main buildings (Hospital and polyclinic) together with other ones distributed around the area for academic purposes.

#### Table 9 - Physical assets of the LetItFlow partners

#### 3.1.3 Human Resources

Table 10 enumerates the different technology and management expertise of the LetItFlow partners.

Partner	Technology and management expertise
Integrasys	Integrasys has a lot of talented persons with technical and managerial experience gained in the last 25 years. The technology background of the Integrasys' staff covers the most advanced and general use technologies and platforms. Integrasys have researchers, software developers, system architects, and product and project managers.



Г



٦

Noldus	Noldus has over 130 employees, most with an academic degree. One of the key resources of Noldus is the domain and application knowledge, which is used for developing products and offering consulting services. Over 1/3 of the staff is involved in R&D, including UI designers, system architects, documentation specialists, and testers.
AIT	AIT's Technology Experience Business Unit can rely on the skills, expertise and knowledge of 30 employees. It is composed of a multidisciplinary research staff (e.g., design, human-computer interaction, computer science, psychology, marketing, sociology, communication sciences, educational sciences, and biology) and deals with user experience (UX) and its various contextual influence factors. The business unit has a long-standing track record specifically in the application of user-centered design approaches with dedicated methods and tools to support the development of high-quality ICT-solutions. The Technology Experience business unit is currently involved in 22 national and international research and development projects, of which it currently coordinates/manages 7.
SIVECO	SIVECO has experience in developing and implementing IT solutions for over 24 years on 4 continents. The company has about 600 employees with experience in national & international projects and with international certifications for various technologies and platforms.
	UHB has managerial resources for international projects, which could manage various teams of physicians in a consortium working on different subjects.
UHB	UHB is also member of the RO-Health cluster. Moreover, in various projects UHB has collaborated with various companies and research institutes such as: Termobit, Medinst, Romanian Medical Technical Institute.
HUVM	HUVM is made up over 5400 employees among doctors, nurses, laboratory technicians, administrative personal, etc. most with an academic degree.

Table 10 – Technology and management expertise

### 3.2 Capabilities

#### 3.2.1 Management of partnerships

Table 10 enumerates the experience of LetItFlow partners on management of partnerships in other projects.

Partner	Management of Partnerships
Integrasys	Integrasys has managerial resources for international projects, which could manage various teams of engineers in a consortium working on different subjects and taking care of different tasks.
Noldus	Noldus plays an active role in a range of leading research projects around the world. In most cases our role is that of a technology partner: we develop tools and methods, and researchers use and validate them. This way we contribute to scientific progress. In the past Noldus has managed several international projects





	and usually we lead one or more work packages within projects.
AIT	As mentioned also above, AIT's Technology Experience Business Unit is currently involved in 22 national and international research projects, where it contributes its expertise mainly in the domain of User Experience (UX) research (e.g., requirements elicitation, user interface design, prototyping, user evaluations). In addition to being a partner, the business unit has a long-standing experience in managing applied research projects. It currently coordinates 7 research projects.
SIVECO	SIVECO has strategic partnerships with international organisations (Intel, ORACLE, Microsoft, IBM, HP, BULL, Ventyx, T-Systems, Samsung, Lattanzio Group, Selex ES, CLICO). The company is involved in international projects as consortium leader or partner and has management specialists with extensive experience in collaborating with remote teams.
UHB	<ul><li>UHB has managerial resources for international projects, which could manage various teams of physicians in a consortium working on different subjects.</li><li>UHB is also member of the RO-Health cluster.</li><li>Moreover, in various projects UHB has collaborated with various companies and research institutes such as: Termobit, Medinst, Romanian Medical Technical Institute.</li></ul>
HUVM	UHB has managerial resources for international projects, which could manage various teams of physicians in a consortium working on different subjects.

#### Table 11 – Management of Partnerships

#### 3.2.2 Sales and Marketing

Table 10 enumerates the experience of LetItFlow partners on marketing products in similar products as LetItFlow.

Partner	Sales and Marketing
Integrasys	Integrasys has no experience on marketing products such as the LetItFlow systems. However, Integrasys has a remarkable product portfolio in the field of signal monitoring and control. The product line of the company is commercialized abroad and thus it has a very talented and experienced sales department, which can apply its experience to other markets and distribution channels.
Noldus	Noldus has a worldwide distribution network, experienced in selling solutions and consultancy projects for behavioral research. Our product portfolio contains a large range of different solutions for behavioral researchers, as such our sales staff is well equipped building relationships with high educated users. Although experiences selling systems comparable to what is being developed within LetItFlow are limited, Noldus is becoming more active and more experienced offering solutions for health applications, for example in the area of health simulation.
AIT	AIT has been involved as a partner in numerous research and development projects, also in the AAL domain, and therefore has repeatedly contributed to the exploitation and dissemination of project results. AIT, however, is typically not





	marketing individual product and service outcomes from project work. As an applied research institution it serves as an important link between basic research (usually at universities) and industry (= technology transfer).
SIVECO	SIVECO has been involved as partner in R&D projects to develop similar IT solutions. The company was not involved though in the Sales & Marketing process. SIVECO has experience in direct sales, in developing dissemination materials and with newsletter campaigns for B2B sector.
UHB	UHB has no experience on marketing products such as the LetItFlow systems.
HUVM	HUVM has no experience on marketing products such as the LetItFlow systems.

Table 12 – Sales and Marketing

# 4. SIMILAR PROJECTS AND PRODUCTS IN THE MARKET

List of similar projects.

PROJECT	Main goals	Year	Website
HELP	The HELP project has designed a system that is able to anticipate when a patient will develop an OFF state or dyskinesia by means of a "Parkinson's" sensor which sends relevant information to the platform and which automatically establishes a new level of drug administration for the pump to overcome that state.	36 Months Starting June 1, 2009	http://www.aal- europe.eu/projects/help/
Inclusion society	The aim of the inclusion society project is to give potential users and customers of technology assisted living solutions the lead by using experience led innovation methods to understand their	36 Months Starting March 1, 2012	http://www.aal- europe.eu/projects/ins/





Γ	noodo from the very		
	needs, from the very beginning and tailor solutions to these.		
Softcare	The SOFTCARE project has developed a prototype of a monitoring system for seniors that allow carers (formal and informal) and senior users to get real-time alarms in dangerous or potentially dangerous situations and warnings on long- term trends that could indicate a future problem.	36 Months Starting November 1, 2009	http://www.aal- europe.eu/projects/softcare/
DOMEO	Focused on the development of an open robotic platform for the integration and adaptation of personalized homecare services, as well as cognitive and physical assistance.	Finished on Dec 31st, 2012	http://www.aal-domeo.org/

#### Table 13 – Similar projects to LetItFlow

List of similar products in the market

PRODUCT	Description	Strong /Weak points	Website
FlowForma	FlowForma is a proven business process enablement tool with the ability to quickly deliver your Everyday, Dark and Collaborative Decision processes with the power of simplicity.	U U	http://www.flowforma.co m/
Cerner Sorian Clinicals	Cerner Soarian Clinicals is a health care information system that offers a revolutionary workflow-driven design	5	http://www.cerner.com/p age.aspx?pageid=1717 9878386&libID=171798 78601





	and addresses the dynamic nature of patient care.		
Critical Alert	Critical Alert Nurse Call Systems creates flexible, highly reliable & secure nurse call systems for hospitals and health care organizations. At the center of our offering is CommonPath, an innovative software application that integrates critical messaging, clinical workflows, real-time location, hardware components and reporting into a single-source solution.	0	http://www.criticalalert.c om

Table 14 – Similar products to LetItFlow

# 5. SWOT ANALYSIS

In this chapter we have carried out a SWOT analysis of the Consortium. SWOT acronym stands for Strengths, Weaknesses, Opportunities, and Threats. SWOT analysis enables a group to move from traditional strategies to a fresh perspective.

Strengths - characteristics of the business that give it an advantage over others in the industry.

# Strengths Technological skills. Experienced and certified specialists. Committed employees. Trust and reciprocity. High quality solution developed by the consortium. Solution tailored for elders needs at work. Value chain covered by consortium.

- Value chain covered by consortium.
- Similar experience with other AAL projects.

#### Table 15 SWOT, strengths

Weaknesses – characteristics that place a firm at a disadvantage relative to others.





#### Weaknesses

- Lack of brand recognition for LetItFlow.
- Lack of a mature solution at this stage.
- Exploitation and market strategies not included in the LIF budget.

#### Table 16 SWOT, weaknesses

Opportunities – chances to make greater profits in the environment. Represent external attractive factors that represent the reason for an organization to exist and develop.

#### Opportunities

- No rival similar solutions dedicated to elder's needs.
- European research and development investments.
- Opportunities in public and private organizations.
- Growing number of older adults with professional knowledge and expertise willing to adopt newer technologies.
- Long term strategies to support active ageing.
- Engagement after retirement of former/actual employees.

#### Table 17 SWOT, opportunities

Threats – external elements in the environment that could trouble for the business. Represent external factors, beyond a organization's control, which could place the organization's mission to operate the risk.

#### Threats

- Rival firms adapt new strategies.
- Entry of foreign competitors.
- Threat of substitution by better solutions.
- Challenge for users' adoption, retention and monetisation.
- Potential reluctance of organisations.

Table 18 SWOT, threats

## 6. INTERVIEWS WITH EXPERTS

A questionnaire is set to tackle new markets is applied on potentials business areas. Thus, in Romania, SIVECO identified potential users for LetItFlow applications in domains with workload and workflows. Examples are health laboratories, security guard and patrol, wastewater treatment plant.





#### Interviews with experts for alternative markets

For the purpose of this project, SIVECO, AIT and HUVM interviewed 42 experts based on the following questionnaire:

#### Questionnaire

The implementation team of the European research and development project **Active Distributed Workflow for Elderly (LetItFLow**) is interested in your opinions concerning the applicability of the IT solutions dedicated to increasing work efficiency and monitoring the staff and resources within an institution/organization.

You will receive the results of this study via e-mail and will be able to use the information in the undertakings regarding optimization of working processes and the activities developed within your the institution/organization.

The **LetItFlow** project aims at designing and implementing specific work flows to support the medical staff aged over 50, i.e. medical nurses, caregivers and laboratory technicians, in fulfilling the daily tasks and aiming to smoothly adjust their work to new procedures, protocols and operation techniques.

1. What field of activity has the institution / organization<sup>5</sup> you represent?

□chemistry / biochemistry;

□nuclear physics / isotopes;

□waste management;

□purification stations;

□geophysics;

□medical (neurology, cardiology, laboratory for medical tests, hematology and blood transfusion);

□guard and protection;

□IT / applied IT.

Are there any persons over 50 years old employed in this institution / organization?
Yes
No

<sup>&</sup>lt;sup>5</sup> Organizational entity of department, compartment, section, unit type etc.
٢	
٦	



Is the activity developed within your institution/organization managed by 3. an information system? □Yes  $\Box$ No (=> MOVE to question no. 8) In case you have answered affirmative to question no. 3, please state 4. whether the existing information system has a software component / application / tool for the management of workflows, processes or activities (for example, BPM -Business Process Management, Task Management)<sup>6</sup>. □Yes □No I do not know (please redirect to a person who might be able to answer this question). Can you please provide details? (such as: implementation location, 5. number of users of the solution, optimized processes and activities, transfer and automatic taking over of responsibilities) Does the existing information system have a software subsystem / 6. component / application / tool for staff monitoring? □Yes □No I do not know (please redirect to a person who might be able to answer this question). Can you please provide details? (such as: video, acoustic, access/ 7. identification card, location, alarm, intelligent equipments/devices) 8. Do you find useful having in your institution an information solution for the management of workflows, processes or activities and/or staff monitoring in order to increase the work efficiency, comfort and security in the workplace of employed persons, especially for those aged over 50? □Yes □No Only for those staff categories that develop their activity under stress factors or in critical/special conditions Delther categories

<sup>&</sup>lt;sup>6</sup> BPM – Business Process Management, Task Management represent specific information solutions for increasing the efficiency and/or monitoring the activity developed within an institutions/organizations





9. Did you undertake any actions in procuring an IT solution / components / applications / software instruments of the type mentioned above?
Yes
No
Do not want to answer
10. Have you received any offer from specialized suppliers? In case your answer is yes, please provide details.
Thank you for your collaboration!
Date: Name \*: Institution/Organization:

#### Table 19 Questionnaire

The interviewed employees have different age and work for the following organisations: University of Veterinary Medicine Vienna, SeneCura Sozialzentrum Grafenwörth, University Municipal Hospital from Bucharest, Horia Hulubei National Institute for R&D in Physics and Nuclear Engineering (IFIN-HH), National Institute for Research and Development in Optoelectronics in Romania, Sibiu Military Hospital, Dinamic Guard Force, Hospital Universitario Virgen Macarena, Duna-Guard and SIVECO Romania. All these organisations are targeted as an alternative market and we detailed that in section *1.5.2. Alternative markets*, of this deliverable.

According to the questionnaire results, 54% of our respondents are working in a medical organization (neurology, cardiology, laboratory for medical test, haematology and blood transfusion) and 34% in a related field, biochemistry as stated in the chart below (Figure 15).







Figure 15 Respondents' domain of activity

In unanimity, all these organizations have employees over 50 years old, as we assumed when we selected the domain for the alternative markets.

Overall, 71.3% of respondents answered that in their organization exists a management information system. The rest who answered "No" are from different domains of activity each. We can speculate that the process of computerizing a business is not strictly related to its activity domain.

Going further, respondents were asked to answer whether the existing information system has a software component / application / tool for the management of workflows, processes or activities as BPM – Business Process Management or Task Management. 43% of them answered "yes" and 22% "don't know" this information. We can assume that over 40% don't have a workflow system because all employees would have been engaged in this process.







Figure 16 Existence of workflow management component in the IT system of the questioned organization

When asked to give extra details about implementation location, number of users of the solution, optimized processes/activities, transfer and automatic taking over of responsibilities, 61% didn't answered while 28% couldn't or didn't knew other details. Only 2 employees from medical and IT domain gave some example, such as: document workflow, access card and Laboratory information management system (LIS). For Laboratory information management system, most equipment is connected into an internal network and each employee has a user for the system. Analysis results are being validated and transmitted through the network.

Asked if the existing information system has a software subsystem / component / application / tool for staff monitoring, 33% answered "No", while 26% have chosen not to answer this question as stated in the chart below (Figure 17).







Figure 17 Staff monitoring tool existence

Moreover, when asked to give examples of monitoring tools such as: video, acoustic, access/ identification card, location, alarm, intelligent equipment /devices, 72.2% have chosen not to answer, others answered "No" and only two respondents (11%) answered "Yes". Among their examples they state: mobile phones, transceiver stations, access card and software modelling.

The next question is directly related to the scope of our solution and mainly to change management. The results state that 67% believe that an information solution for the management of workflows, processes or activities and/or staff monitoring would increase the work efficiency, comfort and security, especially for those aged over 50. From different reasons 28% answered "No" and we can assume that they are not prone to change.







Figure 18 Usefulness of IT solutions for elders

Almost in unanimity, 95% answered that they never undertook any actions in procuring an IT solution / components / applications / software instruments solution for the management of workflows, processes or activities and/or staff monitoring. Taking in account that most of our respondents manage budgets and are involved in decision processes in their activity, we can assume that there is neither budget or there is no interest for this type of solutions.

Related to the previous question, 85% stated that they never received an offer from a specialized IT supplier and the rest chose not to answer.

As a conclusion, we can state that the targeted domains have employees over 50 years old and 50% of them have an Information system with a software component / application / tool for the management of workflows, processes or activities. The most important aspect is related to the willingness of people to adopt a new IT solution for employees over 50 years old. 70% believe that an information solution alike LetItFlow would increase the work efficiency, comfort and security for elders.

# 7. MARKET SEGMENTATION

Market segmentation is the process of dividing consumers into groups based on shared needs, desires and preferences. Normally, a market can be segmented according to four categories: demographics, geography, behaviour and psychographics, where each category can be divided into different groups.

### 7.1 Segmentation by Demography

Demographic market segmentation is one of the most common approaches to segmenting markets based on: age, gender, work state, occupation and income.







### Figure 19 Segmentation by Demography

### Age segmentation

Based on project's scope, we need to divide our market into several segments: people aged between 50 to 60, aged between 60 to 65 and aged larger than 65. Each group could have different requirements regarding products.

### Segmentation by gender

This type of segmentation is relevant because the number of female elderly people is higher than the number of male elderly people and they have different behaviours related to technology. Men are generally more attracted by electronics and computing equipment in comparison with women. Moreover, prediction (see Chapter 2) state that women's involvement in the labour market will increase by 2060.

#### Work state segmentation

Elderly people in our target group are aged more than 50 years and they are either working (fulltime or part-time) or are retired but still have a job. So the market can be segmented along this line: working people and retired people with jobs.





### Occupation by industry segmentation

Is related to our main an alternate markets we target with LetItFlow. Each industry has their own particularities and the people have different specialisations and need different IT tools. According to the targeted market this segment is divided in: chemistry / biochemistry, nuclear physics, waste management, purification stations, geophysics, medical, guard and protection and IT.

### Segmentation by income

Taking into account the fact that the medium wage in Romania (where we have one of the pilots) is around 500 EUR, the first range is beneath <500 EUR. Therefore, more groups can be made.

### 7.2 Geographic segmentation



Geography could also be taken into account from two aspects.

#### rigure zo oeginentation by geogra

### Living in an urban area or in the countryside

People living in cities have different living conditions than people living in village/rural areas. They have different jobs, have better access to technology. In Romania, for example, countryside areas are low developed and in most cases are focused on agriculture and people's interest for technology is significantly decreased.

### **Countries**

Normally, consumer behaviours are diverse in different countries due to local traditions, cultures, policies or economy. Therefore, this factor is important to analyse if we want to sell our products to different countries. We considered non-EU countries separate because of different legislation laws that need to be taken into account.

### 7.3 Behavioural segmentation





Based on desk research and surveys, a lot of behavioural aspects can be considered as segmentation criteria in the project.



Figure 21 Segmentation by behavior

### Health state segmentation

Segmentation by health state is relevant because LetItFlow solution needs to adapt to elder's different disabilities so they can accomplish their daily tasks. This segmentation is divided in: excellent health, good health or poor health.

### Vision and hearing capacity

Regarding vision and hearing capacities we can separate them into good or poor.

### Physical mobility

Physical mobility here means the ability to move and we can separate them into: excellent, good or poor.

### 7.4 Segmentation by Psychographic

Psychographics or lifestyle segmentation and is related to consumers interests and activities.







Figure 22 Segmentation by psychographics

### Attitude to innovation

The surveys for alternative markets revealed that that the interest in technology innovation for elderly support is high. 67% of respondents thought that technology innovations can help elderly to satisfy their needs and problems; 28% of respondents claimed that IT solutions dedicated to elders, as LetItFlow, won't help them improve their work activity. According to these data, the market could be divided into two groups: interested in innovative technologies and not interest in adopting newer technologies.

### IT knowledge

Smart phone, tablets, laptops are common devices that people frequently use for internet access and for different applications. We can divide them into categories: good, satisfactory and insufficient.

LetItFlow has to be accompanied by ongoing training, and long-term technical support. Such support should be an integral part of the package offered. If experiences with the service are positive or expectations are confirmed, people are more likely to use technical services in the future.

# 8. SELECTED SEGMENTATION MARKET

In order to make the segmentation effective, several important and effective segments of project are selected according to the following criteria:

- a. The organization must be able to identify and measure each segment;
- b. The market must be substantial enough;
- c. The organization must be able to reach customers;





- d. Customers in the selected segment must be responsive (money and willingness);
- e. Characteristics of the segment are relatively stable over a long period.

### Assisted group

This target group focuses on the elderly who are limited in some capacity, such as vision, hearing, mobility, etc. LetItFlow shall provide special functions for these people. Older adults experience a lot of restrictions and challenges related to the process of aging. Since visual and cognitive abilities decrease with age, the solution developed within this project could help them in their work productivity.

Assisted group			
Working /Retired but working people over 50 years	Poor Vision or Hearing capacity	Satisfactory or Good IT Knowledge	

#### Table 20 Assisted group

### Productivity increase group

This segmentation aims at the elderly who want to increase their productivity at work. The project will support the elderly in their daily activities by providing applications for their daily tasks. The people in this group are elderly adults who are interested in innovative IT solution, are excited to use them and have satisfactory/good IT Knowledge. In addition, they have an excellent or good health state.

Productivity increase group				
Excellent/Good Health State	Interest in innovative technologies	Satisfactory or Good IT Knowledge		

#### Table 21 Productivity increase group

### Price sensitive group

End-users could choose a mix of functions based on the price and their needs. The economic situation and the number of elderly people in each country should be taken into account, for example, the average income of elderly people in the Romania is lower than that in Spain. Elders in this group are interested in innovative technologies and have satisfactory/good IT knowledge.





Price sensitive group			
Income	Interest in innovative	Satisfactory or Good	
meome	technologies	IT Knowledge	

 Table 22 Price sensitive group

# 9. INITIAL MARKET STRATEGY

This chapter gives a preliminary version of market strategy and business exploitation following the business canvas model approach. A detailed business and exploitation plan will be provided at the end of the project on the deliverable D6.3.

### 9.1 Integrasys

Key Partners National healthcare Services (public and private) Large IT solution provider in Health domain Healthcare network BPM platform providers	Key Activities <u>Development of LetItFlow</u> <u>solution</u> After the end of the project, the developed prototype needs to be mature to become a commercial product. Exhaustive testing in real environment. Need to comply with hospital standards related to privacy, data protection We do not aim an self- installable product, but a customizable platform for the customer needs. In order to be competitive, functions for easy maintenance and remote support need to be developed. <u>Commercial Release</u> LetItFlow as a platform that would be customized	Value Propositions	Customer Relationships Hospital: They are the main customers, since they are the user of the platform. They participate in the requirements stage, to create their customized workflow. Post-sale support is also foreseen. Government: Governments may want to adopt this technology to develop systems that will make it possible for older people to still participate in the working community. Large IT Companies, willing to embedding our workflow solution in their product portfolio.	Customer Segments Hospitals Government
---	--	-----------------------	--	---





	for different workflow. It also is can be integrated with other IT solutions. Key Resources <u>Product development</u> Specialized IT software developer and system integrators, and business process management designer <u>Distribution network</u> Sales team in EU and US of Integrasys. <u>Post-sales team</u> Support team	benefit Introduction of these technologies will have a positive influence on the working situation of elderly nurses and their motivation. This will help to reduce absence and prevent mistakes. This is a benefit for the nurses but also for their employers Usability Interfaces are defined to be usable.	
Cost Structure This describes the most important monetary consequences while operating under different business models <u>Development costs</u> Costs to complete the LetItFlow Tool, including documentation. <u>Communication</u> Personnel costs, plus out of pocket costs (conferences/tradeshows, printing, webhosting, etc.) <u>Sales costs</u> Direct (personnel) for sales, training and support and out of pocket (travel, office space, etc.)		cluding s iosting, etc.)	<ul> <li>Revenue Streams</li> <li>The way a company makes income from each customer segment.</li> <li>Customization of platform for each customer – Implementation cost</li> <li>Post-sale Support – Subscription model</li> </ul>



<u>Running costs</u> Post-sale support



Table 23 Initial Business Plan - Integrasys

# **9.2 SIVECO**

Key Partners	Key Activities	Value	Customer	Customer
National healthcare Services (public & private) Large IT solution provider in Health domain BPM platform providers Own strategic partners (e.g. TSystems for health domain).	Development of LetItFlow In order to have a mature solution SIVECO needs to continue testing the prototype developed during the project. For the solution to be easily customizable, further development might be necessary to have a valid product for the market. Commercial Release	<ul> <li>✔ Development technology ensures the portability on various hardware and software configurations</li> <li>✔ Flexibility to integrate with other systems</li> <li>✔ Tailored to elder's needs and disabilities</li> </ul>	Relationships SIVECO is targeting: own customers, new customers, and also competitors.	Segments Public (B2P) & Private (B2B) Organisations
	LetItFlow would be customizable for different domains.	USP (Unique Selling Propositions)		
	Key Resources Product development and experience in IT for over 24 years. Management and IT experts with know-how and various certifications. Big distribution network with Projects in over 27 countries, across 4 continents.	<ul> <li>LetItFlow: Simple &amp; Intuitive IT solution for elderly</li> <li>Working environment is tailored to elder's needs with LetItFlow</li> <li>LetItFlow is making change easier for the elders</li> </ul>	Channels Communication &PR SIVECO is going to use various tools to disseminate information about LetItFlow solution, such as: press releases, emailing campaigns, telemarketing, social media (FB), own website. The company is also going to participate at fairs, events & conferences to create awareness.	





			Sales & Distribution Assisted by communication tools, SIVECO's sales team, is planning to find potential customers through own customers, partners and own sales network. For both B2B and B2P, the company focuses on direct sales.	
Cost Structure Product management costs Management costs to extend solution's capabilities Development &Testing Personnel cost to improve system's capabilities according to market's needs and to make it easy for customisation. Communication & Sales Costs with dissemination materials, with personnel and for participation at different events to create awareness.		CU	Streams etItFlow solution customizati istomer echnical support post impler	

### Table 24 Initial Business Plan – Siveco

## 9.3 Noldus

Key Partners	Key Activities Development of	Value Propositions	Customer Relationships	Customer Segments	
	LetItFlow	Guidance of nurses	Nurses	Nurses	
	After development	in their daily activities	These are the main end	The LetItFlow tool	
	of the prototype during the project,		users.	can be sold	
	the tool will be	LetItFlow combines solving the problems	Hospitals	directly. To be	
	completed.	of change	Hospitals can adopt this	decided during	
	Commercial	management in companies and	system to support older	the project how to	
	I KEIEdSE I	workflow	staff members and thus	define the actual target group,	
	Release of the	technologies for older nurses by supplying real-time context- aware tools for	increase productivity and	target group, based for	
	LetItFlow Tool as			reduce absence.	example on age,
an independent	an independent product, or as a		ware tools for Governments	acceptance of	
	tool that can be	guiding them in their	Governments may want	technology,	
	embedded in a set	daily tasks. This will increase their	to adopt this technology	educational level.	
	of products and	motivation and	to develop systems that	Hospitals	
	services. This will	independence. Also for their colleagues	will make it possible for	The solution can	
	be done after testing and	this will therefore	older people to still participate in the working	be offered in	
	validation. The full	offer a benefit.	community.	batches to	





product will also include documentation. The release will be supported by a range of PR & amp; communication activities, with the aim to create awareness.	Cost reduction Introduction of these technologies will have a positive influence on the working situation of elderly nurses and their motivation. This will help to reduce absence and prevent mistakes. This is a benefit for the nurses but also for their employers. Easy to learn		hospitals.
Key Resources		Channels	
-	The tool should be	Chamielo	
Data integration	simple to use and	Sales channels	
Experience how to integrate different datastreams into one system, and how to generate information from combining all the available data.	easy to learn.	Partner managers will establish relationships with potential OEM partners for embedding and reselling the LetItFlow tool in solutions they offer.	
Expertise in automatic recognition of behaviour and emotions		Direct sales network can contact governments and hospitals directly.	
using computer vision, physiological data and real-time analysis of interaction logs		Communication & PR In order to create awareness, different tools will be used: newsletters, emailing campaign, social media	
Distribution network		(blogs), website, press releases. In addition we	
Noldus has a worldwide sales network.		can visit relevant conferences and tradeshows to present our tools and get in contact with potential clients and partners. <b>Direct sales to nurses</b>	
		This can be established via a webshop.	





Revenue Streams
Subscription
A subscription model offers the possibility to create a sustainable income flow. Licenses can be sold in batches,
price per batch can depend on batch size and time period.
This model makes it also more affordable for partners who want to use this technology.
Start Price
In order to make sure costs of support during
implementation are covered, there will be a start fee for people who want to start embedding the LetItFlow tool into their application.

### Table 25 Initial Business Plan – Noldus

## 9.4 UHB

Key Partners	Key Activities	Value Propositions	Customer Relationships	Customer
Health clusters; RO-Health	Development of LetItFlow	Improve patients and medical staff	Not relevant for UHB – hence the institution is	Segments Not relevant for UHB hence
Large IT solution provider in Health domain	UHB tests the LetItFlow prototype as an end- user / direct beneficiary	quality of life Strengthens the quality of the	not allowed to sell products.	it is a public institution.
Providers / vendors of medical equipment and	in order to provide quick feedback to the development team and have a reliable solution	medical act. Monitoring and support in critical care.		
devices National healthcare	by the end of the project.	USP (Unique Selling Propositions)		
Services providers (public & private)	LetItFlow will be disseminated in different	✓ LetItFlow: Simple & Intuitive IT		
Pharmaceutical companies Own strategic	domain related workshops and conferences.	solution for elderly ✓ Working		
partners (Medical	Key Resources	environment is tailored to	Channels	
universities)	Key experts in research	elder's needs with LetItFlow	Communication &PR	
	and development projects with and	<ul> <li>✓ LetItFlow is making change</li> </ul>	UHB is going to use various tools to	





	overall background of 20 years. Access to shared knowledge with active international health organisations clusters. Key members in European Boards of different academies/societies.	easie elders	r for the S.	disseminate information about LetItFlow solution by participating in different activities, such as: conferences and congresses. UHB will also contribute to the results of the project in different academic and scientific research papers published in national and international journals.	
Cost Structure		Revenue Streams			
Not applicable		Not applicable			

 Table 26 Initial Business Plan – UHB

# 9.5 HUVM

Key Partners	Key Activities	Value Propositions	Customer Relationships	Customer Segments
Health clusters; Spanish-Health	Development of LetItFlow	Improve patients	Not relevant for HUVM - hence the institution is	Not relevant for HUVM
Large IT solution provider in Health domain	HUVM tests the LetItFlow prototype as an end-user of the project / direct	care, medical and laboratory technician staff quality of life	not allowed to sell products.	hence it is a public institution.
Providers / vendors of laboratory autoanalyze, and clinic analysisbeneficiary in order to provide quick feedback to the development team and have a reliable solution by the end of the project.	Improve care and clinical laboratory diagnostic quality of the hospital	In any case if it is offered as a model for the presentation of the results of the project.		
		Enhance and support monitoring		
reactive, medical	edical laboratory	5		
equipment and devices	LetItFlow will be disseminated in different domain related	technician by medical laboratory. help monitor the work of laboratory monitoring quality		
National healthcare	workshops, conferences and national and international congress of			





Services	clinic laboratory.	control c laborato	of clinical		
providers (public & private)		18001810	ı y		
Pharmaceutical companies.		USP (Ur Selling	nique		
Own strategic partners (Medical universities)		Proposit	,		
		<ul> <li>LetItFlow: Simple &amp; Intuitive IT solution for elderly staff.</li> <li>Working environment is tailored to</li> </ul>			
	Key Resources		s needs _etItFlow	Channels	
	Key experts in research and development	✓ LetItFlow is		Communication &PR	
	projects with and overall background of 30 years.		ng change r for the	HUVM is going to use various tools to disseminate information about LetItFlow solution by participating in different activities, such as: conferences and congresses of clinical laboratory. HUVM will also contribute to the results of the project in different academic and scientific research papers published in national and international journals in collaboration with the different partners of the LetItFlow	
	Access to share all the knowledge and the experience of this project with active international health organisations clusters.	work.			
	Key members in				
	European Boards of different academies/societies, and quality commissions to evaluate the accreditation of the				
	clinical laboratory of the region.			project.	
Cost Structure			Revenue S	Streams	
Not applicable			Not applicable		

### Table 27 Initial Business Plan - HUVM





# **10. CONCLUSIONS**

This document combines both qualitative and quantitative analysis methodologies which provide a good analysis for the market launch and commercial exploitation of the project's results. The methodologies analyse the market from a functional and segmentation point of view. Moreover, this report collects market information and data, providing interesting references for further research and opportunities to be addressed for this project in the future.

Through the external and internal analysis, we now are aware of the opportunities we can benefit from and threats we must care about from a political, economic, social-cultural, technological and even ethical perspective. Besides of the environment, we gained insight into our strengths which could help us compete with our competitors and weaknesses we should overcome. Then it is concluded that our core competence is basically the technology and the innovation. We could provide advanced and innovative technology to help the elderly to do their labour in an independent and intuitive way. Moreover, our strengths could help us to benefit from opportunities better and deal with threats in market. At the same time, we have less experience in marketing and distribution, which is our biggest weakness. Therefore, we should take measures to reinforce this aspect.

Finally, we conclude the document with a basic business plan that will be completed at the end of the project when this project will be more mature and all pending development issues will be finished.





# **11. BIBLIOGRAPHY**

- 1. European Comission, Directorate-General for Economic and Financial Affairs, *The 2015* Ageing Report – Economic and budgetary projections for the 28 EU Member States (2013-2060), European Economy 3/2015.
- 2. AARP Real Possibilities (American Association of Retired Persons www.aarp.org) USA, Health Innovation Frontiers, Updated August 2014.
- 3. Gartner, Predicts 2016: Algorithms Shift the Focus to Business-and Human-Centric Emerging Technologies, Betsy Burton; Mike J. Walker; Alexander Linden; Stephen Prentice; Kenneth F. Brant; Gerald Van Hoy; Gareth Herschel, 03 December 2015 (G00292148) (https://www.gartner.com/doc/3174921/predicts--algorithms-shift-focus).
- 4. Issues in Technology Innovation (Number 18), How Mobile Devices are Transforming Healthcare, Darrell West, May 2012.
- 5. http://ec.europa.eu/eurostat/statisticsexplained/index.php/People\_in\_the\_EU\_%E2%80%93\_statistics\_on\_an\_ageing\_society
- 6. https://ec.europa.eu/research/social-sciences/pdf/policy\_reviews/kina26426enc.pdf
- 7. Short presentation of the University Emergency Hospital Bucharest (history, organization, resources, infrastructure)
- 8. Deliverable D2.1 User needs and requirements
- 9. Website of the University Emergency Hospital Bucharest (UHB) www.suub.ro
- 10. Documents regarding the R&D activity Neurology Department of the University Emergency Hospital Bucharest.
- 11. Eurostat statistics http://ec.europa.eu/eurostat/help/new-eurostat-website
- 12. Europop2013 (European Population Projections, base year 2013) http://ec.europa.eu/eurostat/web/population-demography-migration-projections/populationprojections-data
- 13. Technology for Adaptive Aging, written by Steering Committee for the Workshop on Technology for Adaptive Aging, Board on Behavioral, Cognitive, and Sensory Sciences, Division of Behavioral and Social Sciences and Education, National Research Council.
- 14. Smartphone user penetration as percentage of total population in Western Europe from 2011 to 2018\*. http://www.statista.com/statistics/203722/smartphone-penetration-per-capita-in-western-europe-since-2000/